Author’s response to reviews

Title: MyoMiner: Explore Gene Co-expression in Normal and Pathological Muscle

Authors:

Apostolos Malatras (apmalatras@biol.uoa.gr)
Ioannis Michalopoulos (imichalop@bioacademy.gr)
Stephanie Duguez (s.duguez@ulster.ac.uk)
Gillian Butler-Browne (gillian.butler-browne@upmc.fr)
Simone Spuler (Simone.Spuler@charite.de)
William J Duddy (w.duddy@ulster.ac.uk; billduddy@hotmail.com)

Version: 6 Date: 07 Apr 2020

Author’s response to reviews:

Dear editor,

All of the supp tables are now referenced in the main text.

Regarding the abstract, can you please specify what the problem is with it? We previously followed the instructions in your email of 26/03:

FORMATTING CHANGES:

The abstract must include the following separate sections:

Background: the context and purpose of the study

Methods: how the study was performed and statistical tests used

Results: the main findings

Conclusions: brief summary and potential implications